Author's response to reviews

Title: Chlamydia trachomatis genovar distribution in clinical urogenital specimens from Tunisian patients: high prevalence of C. trachomatis genovar E and mixed infections

Authors:

Houda GHARSALLAH (houda_gharsallah@yahoo.fr)
Olfa FRIKHA GARGOURI (olfafrikhagargouri@ymail.com)
Hanen SELLAMI (sellami.hanen@yahoo.fr)
Fatma BESBES (besbes.fatma@gmail.com)
Abir ZNAZEN (abirznazen2001@yahoo.fr)
Adnene HAMMAMI (adnene_hammami@rns.tn)

Version: 5 Date: 15 October 2012

Author's response to reviews: see over
Dear Professor,

Thank you very much for your e-mail dealing with our manuscript sent for publication in BMC infectious Disease (Ms ID: 1458862786701781). Thanking you for your kind invitation to resubmit our article after making revision. This version has been revised according to the reviewers’ recommendations. All specific requests are now changed and included in the text. The English of the manuscript has been thoroughly checked.

All modifications included in the revised version, are marked in red.

Please find enclosed point-by-point answers to reviewers’ comments and the revised manuscript that we hope acceptable for publication in BMC infectious Disease as regular paper.

rThanking you

Yours sincerely

PhD. Houda Gharsallah